Author's response to reviews

Title: Social determinants of rest deprivation amongst Ghanaian women: national and urban-rural comparisons with data from a cross-sectional nationally representative survey

Authors:

Maurice B Mittelmark (maurice.mittelmark@uib.no)
Torill Bull (torill.bull@uib.no)

Version: 6 Date: 28 September 2010

Author's response to reviews:

A few adjustments were made to clarify the text. No substantive changes were made.